Author’s response to reviews

Title: Roles and effectiveness of lay community health workers in the prevention of mental, neurological and substance use disorders in low and middle income countries: A Systematic Review

Authors:

Byamah B Mutamba (byamamutamba@yahoo.com)
Nadja vanGinneken (nadja.vanginneken@lshtm.ac.uk)
Lucy P Smith (Lucy.Paintain@lshtm.ac.uk)
Symon Wandiembe (symon.wandiembe@lshtm.ac.uk)
David Schellenberg (David.Schellenberg@lshtm.ac.uk)

Version: 2 Date: 20 October 2012

Author’s response to reviews:

Dear Editorial team, the corresponding author has made changes to his professional title, from Professor to Doctor,

Thank you